Author's response to reviews

Title: Prevalence and types of thyroid malignancies among thyroid enlarged patients in Gondar, Northwest Ethiopia: a three years institution based retrospective study

Authors:

  Tadele Melak Mr. (0923tadie@gmail.com)
  Biniam Mathewos Mr. (fikirbinny@gmail.com)
  Bamlaku Enawgaw Mr. (bamlaku.enawgaw@gmail.com)
  Debasu Damtie Mr. (debidam@gmail.com)

Version: 4 Date: 15 February 2014

Author's response to reviews: see over
To: BMC cancer

We, the author of the research article entitled “Prevalence and types of thyroid malignancies among thyroid enlarged patients in Gondar, Northwest Ethiopia: a three years institution based retrospective study” have decided to submit our research article to BMC cancer journal because of the fact that our work has direct relevance to the journal. Moreover, we believe that the article would add something to the scientific community when published on this journal. We would like also to declare that we don’t have any competing interest about the research article.